Author's response to reviews

Title: Management of major bile duct injury after laparoscopic cholecystectomy:
A case report and review of the literature

Authors:

Andreas Manouras (amanouras@hippocratio.gr)
Nikolaos Pararas (npararas@hotmail.com)
Pantelis Antonakis (antonakis_pantelis@yahoo.gr)
Emanuel E. Lagoudianakis (redemlag@yahoo.gr)
George Papageorgiou (awace@yahoo.gr)
Ioannis G. Dalianoudis (johndalas@gmail.com)
Manousos M. Konstadoulakis (labsures@hippocratio.gr)
Vaggelogiannis Katergianakis (vkater@hotmail.com)

Version: 2 Date: 17 June 2008

Author's response to reviews:

We sent this revision of the article after your email. We have made the article
with your instructions in order to conform with your journal style. We have
removed the keywords and included a list of the abbreviations. We have included the
informations about the ethnicity of the patient and the consent form as well the
competing interest section and the authors contribution section.